\hrefanchor
\outdef{citeabeSeO}{ABe71}
\hbox{\parbox[t]{\rw}{[\cite{abeSeO}]}\parbox[t]{\colw}{D.~Abraham, E.~Barouch, G.~Gallavotti, A.~Martin-L\"of - {\it Dynamics of a local perturbation in the XY model - I.~Approach to equilibrium}, Studies in Applied Mathematics, Vol.~50, p.~121-131, 1971.}}\par
\bigskip

\hrefanchor
\outdef{citeandSeZ}{And70}
\hbox{\parbox[t]{\rw}{[\cite{andSeZ}]}\parbox[t]{\colw}{P.~Anderson - {\it A poor man's derivation of scaling laws for the Kondo problem}, Journal of Physics C: Solid State Physics, Vol.~3, p.~2436, 1970.}}\par
\bigskip

\hrefanchor
\outdef{citeandSO}{And61}
\hbox{\parbox[t]{\rw}{[\cite{andSO}]}\parbox[t]{\colw}{P.~Anderson - {\it Localized magnetic states in metals}, Physical Review, Vol.~124, n.~1, p.~41-53, 1961.}}\par
\bigskip

\hrefanchor
\outdef{citeayhSeZ}{AYH70}
\hbox{\parbox[t]{\rw}{[\cite{ayhSeZ}]}\parbox[t]{\colw}{P.~Anderson, G.~Yuval, D.~Hamann - {\it Exact results in the Kondo problem - II. Scaling theory, qualitatively correct solution, and some new results on one-dimensional classical statistical mechanics}, Physical Review B, Vol.~1, n.~11, p.~4464-4473, 1970.}}\par
\bigskip

\hrefanchor
\outdef{citeaySN}{AY69}
\hbox{\parbox[t]{\rw}{[\cite{aySN}]}\parbox[t]{\colw}{P.~Anderson, G.~Yuval - {\it Exact results in the Kondo problem: equivalence to a classical one-dimensional Coulomb gas}, Physical Review Letters, Vol.~23, n.~2, p.~89-92, 1969.}}\par
\bigskip

\hrefanchor
\outdef{citeandEZ}{And80}
\hbox{\parbox[t]{\rw}{[\cite{andEZ}]}\parbox[t]{\colw}{N.~Andrei - {\it Diagonalization of the Kondo Hamiltonian}, Physical Review Letters, Vol.~45, n.~5, 1980.}}\par
\bigskip

\hrefanchor
\outdef{citeaflETh}{AFL83}
\hbox{\parbox[t]{\rw}{[\cite{aflETh}]}\parbox[t]{\colw}{N.~Andrei, K.~Furuya, J.~Lowenstein - {\it Solution of the Kondo problem}, Reviews of Modern Physics, Vol.~55, n.~2, p.~331-402, 1983.}}\par
\bigskip

\hrefanchor
\outdef{citebgNZ}{BG90}
\hbox{\parbox[t]{\rw}{[\cite{bgNZ}]}\parbox[t]{\colw}{G.~Benfatto, G.~Gallavotti - {\it Perturbation theory of the Fermi surface in a quantum Liquid - a general quasiparticle formalism and one-dimensional systems}, Journal of Statistical Physics, Vol.~59, n.~2-3, p.~541-664, 1990.}}\par
\bigskip

\hrefanchor
\outdef{citebgeNF}{BGe90}
\hbox{\parbox[t]{\rw}{[\cite{bgeNF}]}\parbox[t]{\colw}{G.~Benfatto, G.~Gallavotti, A.~Procacci, B.~Scoppola - {\it Beta function and Schwinger functions for a many Fermions system in one dimension - Anomaly of the Fermi surface}, Communications in Mathematical Physics, Vol.~160, p.~93-171, 1994.}}\par
\bigskip

\hrefanchor
\outdef{citedorNO}{Dor91}
\hbox{\parbox[t]{\rw}{[\cite{dorNO}]}\parbox[t]{\colw}{T.~Dorlas - {\it Renormalization group analysis of a simple hierarchical fermion model}, Communications in Mathematical Physics, Vol.~136, p.~169-194, 1991.}}\par
\bigskip

\hrefanchor
\outdef{citedysSN}{Dys69}
\hbox{\parbox[t]{\rw}{[\cite{dysSN}]}\parbox[t]{\colw}{F.~Dyson - {\it Existence of a phase-transition in a one-dimensional Ising ferromagnet}, Communications in Mathematical Physics, Vol.~12, p.~91-107, 1969.}}\par
\bigskip

\hrefanchor
\outdef{citekitSeS}{Kit76}
\hbox{\parbox[t]{\rw}{[\cite{kitSeS}]}\parbox[t]{\colw}{C.~Kittel - {\it Introduction to solid state physics}, Wiley\&Sons, 1976.}}\par
\bigskip

\hrefanchor
\outdef{citekonSF}{Kon64}
\hbox{\parbox[t]{\rw}{[\cite{konSF}]}\parbox[t]{\colw}{J.~Kondo - {\it Resistance minimum in dilute magnetic alloys}, Progress of Theoretical Physics, Vol.~32, n.~1, 1964.}}\par
\bigskip

\hrefanchor
\outdef{citekonZFi}{Kon05}
\hbox{\parbox[t]{\rw}{[\cite{konZFi}]}\parbox[t]{\colw}{J.~Kondo - {\it Sticking to my bush}, Journal of the Physical Society of Japan, Vol.~74, n.~1, p.~1-3, 2005.}}\par
\bigskip

\hrefanchor
\outdef{citenozSeF}{Noz74}
\hbox{\parbox[t]{\rw}{[\cite{nozSeF}]}\parbox[t]{\colw}{P.~Nozi\`eres - {\it A ``Fermi-liquid'' description of the Kondo problem at low temperatures}, Journal of Low Temperature Physics, Vol.~17, n.~1-2, p.~31-42, 1974.}}\par
\bigskip

\hrefanchor
\outdef{citerueSN}{Rue69}
\hbox{\parbox[t]{\rw}{[\cite{rueSN}]}\parbox[t]{\colw}{D.~Ruelle - {\it Statistical mechanics: rigorous results}, Imperial College Press, World Scientific, first edition: Benjamin, 1969, 1999.}}\par
\bigskip

\hrefanchor
\outdef{citeshaNF}{Sha94}
\hbox{\parbox[t]{\rw}{[\cite{shaNF}]}\parbox[t]{\colw}{R.~Shankar - {\it Renormalization group approach to interacting fermions}, Reviews of Modern Physics, Vol.~66, n.~1, p.~129-192, 1994.}}\par
\bigskip

\hrefanchor
\outdef{citewilSeFi}{Wil75}
\hbox{\parbox[t]{\rw}{[\cite{wilSeFi}]}\parbox[t]{\colw}{K.~Wilson - {\it The renormalization group: Critical phenomena and the Kondo problem}, Reviews of Modern Physics, Vol.~47, n.~4, 1975.}}\par
\bigskip

\hrefanchor
\outdef{citewilSFi}{Wil65}
\hbox{\parbox[t]{\rw}{[\cite{wilSFi}]}\parbox[t]{\colw}{K.~Wilson - {\it Model Hamiltonians for local quantum field theory}, Physical Review, Vol.~140, n.~2B, p.~445-457, 1965.}}\par
\bigskip

\hrefanchor
\outdef{citewilSeZ}{Wil70}
\hbox{\parbox[t]{\rw}{[\cite{wilSeZ}]}\parbox[t]{\colw}{K.~Wilson - {\it Model of coupling-constant renormalization}, Physical Review D, Vol.~2, n.~8, p.~1438-1472, 1970.}}\par
\bigskip

